Author’s response to reviews

Title: Understanding resilience in same-sex parented families: the work, love, play study

Authors:

Jennifer Power (jennifer.power@latrobe.edu.au)
Amaryll Perlesz (A.Perlesz@latrobe.edu.au)
Schofield J Margot (M.Schofield@latrobe.edu.au)
Pitts Marian (M.Pitts@latrobe.edu.au)
Brown Rhonda (rhonda.brown@deakin.edu.au)
McNair Ruth (r.mcnaire@unimelb.edu.au)
Barrett Anna (barrett.anna@gmail.com)
Bickerdike Andrew (abickerdike@rav.org.au)

Version: 3 Date: 5 March 2010

Author’s response to reviews:

Dear Editor,

Please find attached a revised version of our study protocol paper titled, “Understanding resilience in same-sex parented families: the Work, Love, Play study”.

The major and minor revisions have been completed as requested. This includes: correcting the style of the author list, removing the text ‘study protocol paper’ from the title page, correcting the title case, including email addresses for all authors on the title page, correcting references as noted, renaming the methods section ‘methods/design’ and removing visible vertical lines from the table. The manuscript has also been proofread again for typographical errors.

If you require more information, I can be contacted via email to Jennifer.power@latrobe.edu.au or phone to +61 3 9385 5100.

Best regards

Jennifer Power